Title: Counselling versus low-intensity cognitive behavioural therapy for persistent sub-threshold and mild depression (CLICD): A pilot/feasibility randomised controlled trial

Authors:

Elizabeth Freire (bethfrei@gmail.com)
Christopher Williams (chris.williams@glasgow.ac.uk)
Claudia-Martina Messow (martina.messow@glasgow.ac.uk)
Mick Cooper (mick.cooper@roehampton.ac.uk)
Robert Elliott (robert.elliott@strath.ac.uk)
Alex McConnachie (alex.mcconnachie@glasgow.ac.uk)
Andrew Walker (Andrew.walker@glasgow.ac.uk)
Deborah Heard (deborah.heard@gmail.com)
Jill Morrison (jill.morrison@glasgow.ac.uk)

Version: 3
Date: 5 February 2015

Author's response to reviews: see over
February 5th, 2015

Dear Dr. Shipley:

Re: Counselling versus Low-Intensity Cognitive behavioural therapy for persistent sub-threshold and mild Depression (CLICD): A pilot/feasibility randomised controlled trial
Manuscript number: 4946568931592218

Please find attached a revised version of our manuscript which now has the line and page numbers inserted as requested. We apologise for this omission in our original submission.

Yours sincerely,

Professor Jill Morrison
Institute of Health and Wellbeing
College of Medical, Veterinary and Life Sciences
University of Glasgow
1 Horselethill Road, Glasgow G12 9LX
Tel: +44 (0)141 330 8348 Fax: +44 (0)141 330 8331
e-mail: Jill.morrison@glasgow.ac.uk